Author's response to reviews

Title: Pathways into homelessness: recently homeless adults - problems and service use before and after becoming homeless in Amsterdam

Authors:

Igor RAL van Laere (ivlaere@ggd.amsterdam.nl)
Matty AS de Wit (Mdwit@ggd.amsterdam.nl)
Niek S Klazinga (N.S.Klazinga@amc.uva.nl)

Version: 5 Date: 4 December 2008

Author's response to reviews: see over
Dear Editor,

The formatting changes as requested below have been made accordingly.

Formatting changes requested
---------------------------------
References - References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example 'text [1].'

Yours sincerely,

Igor van Laere, Matty de Wit and Niek Klazinga

Amsterdam